\documentclass{article}
\usepackage{cs861}

\begin{document}
	
\lecturetitle{1}{Lecture 01: PAC Learning}{Scribe 1 name, Scribe 2 name}{09/06/2023}

% Include this notice at the beginning of the notes ==================================
\paragraph{Disclaimer:}
\emph{These notes have not been subjected to the usual scrutiny
reserved for formal publications.
They may be distributed outside this class only with the permission of the
instructor.}
\vspace{0.2in}

% Provide a brief synopsis of what we did in the lecture =============================
In this lecture, we will introduce PAC Learning.
We will first introduce \texttt{Topic 1}, then discuss \texttt{Topic 2}, 
and finally conclude with \texttt{Topic 3}.


\section{Topic 1}

Separate each topic we covered in class into different sections.

\section{Topic 2}

\section{Topic 3}



\end{document}
